Answering Reviewers

1. **Time after the last menstruation period**
   The patient stated that her last menstrual period was 13 days before admission. I have added this information in the part of “History of present illness” in revised manuscript.

2. **The patient had no history of pelvic inflammatory disease or pelvic surgery.**
   I have added this information in the part of “History of past illness” in revised manuscript.

3. **In the histopathology section, the authors should clarify that other differentials were excluded including hydatidiform mole.**
   Histopathological investigation revealed chorionic villi within the mass, with no features of abnormal trophoblastic proliferation, which confirming the diagnosis of ectopic pregnancy and excluding other differentials.